Author's response to reviews

Title: The GABBR1 locus and the G1465A variant is not associated with temporal lobe epilepsy preceded by febrile seizures.

Authors:

Shaochun Ma (shaochun.ma@vanderbil.edu)
Bassel Abou-Khalil (bassel.abou-khalil@vanderbilt.edu)
James S Sutcliffe (james.s.sutcliffe@vanderbilt.edu)
Jonathan L Haines (Jonathan@chgr.mc.vanderbilt.edu)
Peter Hedera (peter.hedera@vanderbilt.edu)

Version: 2  Date: 25 March 2005

Author's response to reviews:

Ad reviewer #1: (Minor essential revisions)
None
Ad reviewer #1: (Discretionary revisions)
1. Last sentence of the abstract was divided into two sentences as suggested (Abstract, Conclusions)
2. We added one paragraph (Discussion, page 10, 3rd paragraph) discussing possible explanations of our failure to replicate the original study.
3. The abstract was shortened and now is one page long.

Ad reviewer #2 (Minor essential revisions)
The manuscript was carefully edited and all grammatical mistakes corrected.
Ad reviewer #2: (Discretionary revisions)
1. We added the description of MRI methodology (Methods, page 4, 1st paragraph)